Author's response to reviews

Title: Oral rehydration versus intravenous therapy for treating dehydration due to gastroenteritis in children: a meta-analysis of randomised controlled trials

Authors:
Steven Bellemare (sb6@ualberta.ca)
Ms Lisa Hartling (hartling@ualberta.ca)
Ms Natasha Wiebe (nwiebe@ualberta.ca)
Kelly Russell (krussell@ualberta.ca)
William R Craig (wcraig@cha.ab.ca)
Don McConnell (DMcConne@cha.ab.ca)
Dr Terry P Klassen (terry.klassen@ualberta.ca)

Version: 6 Date: 5 Apr 2004

See my responses to your queries in [brackets].

Queries:
* Page 2, line 19: please define RD. [Done]
* Page 2, line 21: '95% CI' added here. Please confirm that this is correct? [OK]
* Page 2, line 26: Please define WMD. [Done]
* Page 2, line 26: Are the figures in brackets (-2.4,-0.02) 95% CI? [yes]
* Page 2, line 28: Are the figures in brackets (20, 100) 95% CI? [yes]
* Page 5, line 24: Please define RD here. [done]
* Page 5, line 27: To what does the 4% refer? [revised]
* Page 6, line 13: Table 4 is not provided. Please supply the table or correct the citation. [done]
* Page 7, line 49: Please define ORS. [done]
* Page 8, line 2: Please define OR. [done]
* Figure legends: Please define RD. [done]
* Figure 2: Please change wording to sentence case, and change 'favour' to 'favor'. [done]
* Figure 3: Please change wording to sentence case. [done]
* Tables: Please reformat your tables so that they fit onto the page in portrait orientation. If this is not possible, please upload them as additional files, changing their citation in the text accordingly. [these have been uploaded as additional files]